Author's response to reviews

Title: Effectiveness of a MF-59TM-adjuvanted pandemic influenza vaccine to prevent 2009 influenza A/H1N1-related hospitalisation; a matched case-control study

Authors:

Anneke Steens (anneke.steens@rivm.nl)
Eleonora G Wijnans (e.wijnans@erasmusmc.nl)
Jeanne P Dieleman (j.dieleman@erasmusmc.nl)
Miriam CJM Sturkenboom (m.sturkenboom@erasmusmc.nl)
Marianne AB van der Sande (marianne.van.der.sande@rivm.nl)
Wim van der Hoek (wim.van.der.hoek@rivm.nl)

Version: 4 Date: 14 July 2011

Author's response to reviews:

Dear editor,

Please find attached our accepted manuscript ‘Effectiveness of a MF-59TM-adjuvanted pandemic influenza vaccine to prevent 2009 influenza A/H1N1-related hospitalisation; a matched case-control study’ with manuscript number MS 151586902508337. In this version we have made some minor textual changes.

Yours sincerely,

Wim van der Hoek